

IFAU – INSTITUTE FOR LABOUR MARKET POLICY EVALUATION

The importance of education for the reallocation of labor: evidence from Swedish linked employer-employee data 1986–2002

Marie Gartell Ann-Christin Jans Helena Persson

WORKING PAPER 2007:14

The Institute for Labour Market Policy Evaluation (IFAU) is a research institute under the Swedish Ministry of Employment, situated in Uppsala. IFAU's objective is to promote, support and carry out: evaluations of the effects of labour market policies, studies of the functioning of the labour market and evaluations of the labour market effects of measures within the educational system. Besides research, IFAU also works on: spreading knowledge about the activities of the institute through publications, seminars, courses, workshops and conferences; influencing the collection of data and making data easily available to researchers all over the country.

IFAU also provides funding for research projects within its areas of interest. The deadline for applications is October 1 each year. Since the researchers at IFAU are mainly economists, researchers from other disciplines are encouraged to apply for funding.

IFAU is run by a Director-General. The authority has a board, consisting of a chairman, the Director-General and seven other members. The tasks of the board are, among other things, to make decisions about external grants and give its views on the activities at IFAU. A reference group including representatives for employers and employees as well as the ministries and authorities concerned is also connected to the institute.

Postal address: P.O. Box 513, 751 20 Uppsala Visiting address: Kyrkogårdsgatan 6, Uppsala Phone: +46 18 471 70 70 Fax: +46 18 471 70 71 ifau@ifau.uu.se www.ifau.se

Papers published in the Working Paper Series should, according to the IFAU policy, have been discussed at seminars held at IFAU and at least one other academic forum, and have been read by one external and one internal referee. They need not, however, have undergone the standard scrutiny for publication in a scientific journal. The purpose of the Working Paper Series is to provide a factual basis for public policy and the public policy discussion.

ISSN 1651-1166

The importance of education for the reallocation of labor: evidence from Swedish linked employeremployee data 1986–2002*

Marie Gartell**, Ann-Christin Jans[¤], Helena Persson[#]

May 8, 2007

Abstract

Using employer-employee data covering the whole Swedish economy over a uniquely long time period from 1986 to 2002, we examine how job flows and worker flows have been distributed both on an aggregate level and across educational levels. We find that job and worker flows vary by educational level, not only with respect to magnitude and variation, but with respect to direction as well. Our results show that analyses that do not account for the educational level of workers can be very misleading.

Keywords: Linked employer-employee data, job and worker flows, education JEL-codes: J23, J21, I20

**Institute for futures studies (IF), Box 591, 101 31 Stockholm, E-mail:

^{*}Data and financial support for Gartell and Jans from the Institute for Labour Market Policy Evaluation (IFAU) is gratefully acknowledged. Persson thanks SACO for allowing time for the project. We would also like to thank SACO for the financing of the english reviewing. We are grateful to Gunnar Isacsson, Thomas Lindh, Oskar Nordström Skans and Erik Mellander for valuable comments. We have also benefited from comments made by participants at the EALE conference in Prague 2006 and at seminars at IFAU, IF, the Swedish Institute for Social Research and Statistics Sweden.

marie.gartell@framtidsstudier.se

^a Swedish Social Insurance Agency, Department of Research, Analysis and Statistics, 103 51 Stockholm, Sweden, E-mail: ann-christin.jans@socialagency.se

[#]Swedish Confederation of Professional Associations (SACO), Box 2206, 103 15 Stockholm, Sweden, E-mail: helena.persson@saco.se

Table of contents

1	Introduction	3
2	Characteristics of the Swedish labor market	5
3	Data	7
4	Job flows and worker flows in Sweden 1986 to 2002	9
4.14.24.3	The concept of Job flows The concept of worker flows Flows across groups	10 11 13
5 5.1 5.2	Estimated flows for the whole population Job flows Worker flows	14 14 18
6 6.1 6.2	Job flows and worker flows for different educational groups Job flows Worker flows	21 22 25
7	Education and the long-run reallocation of labor	
8	Conclusions	
Refe	erences	
APP	'ENDIX	

1 Introduction

There has been a proliferation of literature on job and worker flows in recent years.¹ Evidence from several countries has shown that firms and workers are heterogeneous and that a great deal of job creation and destruction, as well as hirings and separations of workers, occurs simultaneously. The flows of jobs and workers give evidence of the complexity of the dynamics on the labor market even when net employment does not alter much.

A shortcoming of most previous studies is that they cover short periods of time, and/or are limited to only one or few industries in the economy, mostly manufacturing. Although the heterogeneity of establishments is often taken into account, workers are generally treated as homogeneous. However, analyses of job and worker flows that do not account for the heterogeneity of workers can be very misleading. We contribute to previous literature by studying a very long period from 1986 to 2002, covering both upturns and downturns in the economy. Moreover, this study considers how job and worker flows are distributed for the whole of the Swedish economy.² We find that job and worker flows are substantial across years, and with huge variations. In addition, we decompose the workers into educational groups, and demonstrate that job and worker flows vary by educational level, not only with respect to magnitude and variation, but with respect to direction as well. Further, the volume of worker flows is more than twice as high as that needed to match job flows, and increases with educational level. Thus, much of the mobility on the labor market cannot be explained by the reallocation of jobs.

Due to the long period studied, and the widely fluctuating business cycle during this period, we have had a unique opportunity to analyze the cyclical pattern of job and worker flows. The question of whether or not job and worker flows are countercyclical has been discussed in many previous studies. If a countercyclical pattern has been found, it has often been explained by recessions being periods of intense restructuring activity in the economy. A number of models have been developed to incorporate the cyclical pattern of job and worker flows. Mortensen & Pissarides (1994) present a matching model of unemployment with endogenous job creation

¹ See Abowd & Kramarz (1999) and Davis & Haltiwanger (1999) for overviews of studies using linked employer-employee data on job and worker flows. Flows of workers are presented in e.g. Hamermesh et al. (1994), Lane et al. (1996), Belzil (1997), Albæk & Sørenssen (1998), Abowd et al. (1999) and Salvanes & Førre (2003). See Andersson et al. (1998), Persson (1999), Arai & Heyman (2000), Andersson (2003) and Nordström Skans et al. (2006) for studies using Swedish data.

² Persson (1999) also covers the whole economy, but for a shorter time period (1986–1995).

and job destruction processes. During upturns it takes time to fill vacancies while during downturns, job destruction occurs immediately. Job turnover is thus countercyclical. Garibaldi (1998) extends the Mortensen & Pissarides model by allowing for employment protection legislation in the form of fixed firing costs. When firing is costly and time-consuming, the asymmetry in the cyclical pattern of job creation and job destruction disappears and job destruction becomes less responsive. Higher firing costs can result in acyclical or even procyclical movement of job reallocation (both acyclical and procyclical patterns have been found for some countries). When workers are treated as homogenous, we find that job reallocation in Sweden is countercyclical, which, according to Garibaldi, would suggest that firing costs are quite low. However, when we estimate correlations for different educational groups, we find job reallocation to be countercyclical for the lowest educated, acyclical for the medium educated and procyclical for the highest educated workers.

Previous studies on job and worker flows for the Swedish labor market have showed that job reallocation rates were of the same order of magnitude in 1995 as in the mid 1980s while worker reallocation rates had declined by more than 15 percent. Due to the short period covered, it was not possible, however, to tell whether this was an effect of the severe downturn in the beginning of the 1990s or whether it was a structural change towards lower levels of worker mobility. When examining job reallocation rates for a considerably longer period (1986–2002) we find job reallocation rates to be stable both in order and magnitude.³ Turning to worker flows, we find that worker reallocation rates have exhibited a more volatile pattern, but without a trend and consequently find no support for decreased worker mobility in the Swedish labor market. In addition, we find (confirming previous results) that both job and worker flows are dominated by flows corresponding to existing establishments, while job creation due to new establishments and job destruction due to exiting establishments constitute a minor part.

In Sweden and many other industrialized countries, the last decades of the 20th century have been characterized by a strong relative shift in employment towards more highly-educated employees. In the same period, the supply of highly-educated workers increased while the labor market situation deteriorated for less educated workers. As mentioned above, job and worker flows exhibit clear cyclical patterns. However, we also need to study the long-run, structural pattern of flows

³ Contrary to our findings, Davis et al. (2005) find that job reallocation rates in the U.S. have experienced a downward trend in recent decades.

to increase the understanding of the underlying mechanisms responsible for the change in relative employment for different educational groups. A question previously addressed by Salvanes & Førre (2003) for Norway is whether it is supply or demand that is the driving force behind the change. The restructuring that went on during the 1990s could have been driven either by the birth of "new" jobs, where employers demanded new skills and assignments (demand effects), or by "old" jobs filled merely by more highly educated workers (supply effects), or by a combination of both. Our results indicate that the demand effect is important, but also that the increased supply of higher educated workers has been essential for the changes on the Swedish labor market.

We have structured the paper as follows: Section 2 presents a brief overview of the Swedish labor market, and Section 3 describes the data. Section 4 includes the definitions of job and worker flows. Section 5 presents the estimated flows for the whole population and Section 6 presents the estimated flows for educational groups. In Section 7, the long-run reallocation for different educational groups is investigated. The paper ends with conclusions in Section 8.

2 Characteristics of the Swedish labor market

Here follows a brief overview of the Swedish labor market to serve as a background to our findings. As in most other countries, the Swedish labor market has experienced a decrease in the supply of less educated workers. *Figure 1* shows that the supply of workers educated only to pre-upper secondary level, as a share of the total labor force, declined from 39 percent in 1985 to 22 percent in 2003. Conversely, the supply of university-educated workers increased from 15 percent in 1985 to 30 percent in 2003.



Figure 1. Educational distribution in the labor force among those aged 16–64 years, 1985–2003

Source: AKU, Statistics Sweden.

Our study covers the period 1986-2002, which includes both upturns as well as downturns in the economy. Indeed, the period covers the deepest recession in Sweden since the Great Depression in the 1930s. Employment increased during the late 1980s and peaked in 1990 with more than 4.5 million employees. This period was followed by a deep economic downturn and a decline to 3.9 million employees at the end of the 90s. Although employment has recovered since 1997, employment is still lower than in 1987. Between 1987 and 2003, the total employment figure decreased by around 80 000. Unemployment increased from very low levels, from around 1.5 percent at the end of the 1980s to over 8 percent in 1993. At the beginning of the 2000s unemployment had fallen to around 4–5 percent.

The decline in employment has, however, not been evenly distributed across educational groups. Relative employment for different groups is presented in the left-hand section of *Figure 2*. People with the lowest levels of education have also had the lowest rate of employment throughout the period. Moreover, employment rates have declined for every educational group between 1987 and 2003, but the decrease has been most pronounced for the least educated group.

Figure 2. Employment and unemployment rates for different educational and age groups, 1987–2003



Source: AKU, Statistics Sweden.

Note: Between 1995 and 1996, of those outside of the labor force, 70 percent of the group with no defined education went on to receive a defined education. These people are therefore found in the other educational groups from 1996 and onwards. As a result, employment rates decreased in 1996 since the denominator then became larger for each educational group.

As is clear from *Figure 1*, the supply of lesser-educated people has fallen. Even so, from 1992 and onwards, unemployment for the less-educated has been higher than for all other educational groups. The unemployment rates for different groups is presented in the right-hand section of *Figure 2*.

3 Data

Annual employer-employee linked data from IFAU/Statistics Sweden has been used to study job flows and worker flows. The basic observational unit underlying job and worker flows is the establishment: the physical location where the production takes place. A company or firm is a legal economic entity that encompasses one or more establishments.

The Swedish Employment Register contains the whole population aged 16 or above, in November each year. The connection to the employer of all those employed or self-employed in November is denoted by the identity numbers of the firm and the establishment where each individual had his or her main work. These identity numbers are taken from the Business Register (CFAR).⁴ Establishment level data contains information on its geographical location, industry, total number of employees and form of ownership. In addition to information on the employer and the establishment, individual level data contains detailed information on various individual specific variables such as education, age, country of birth, annual earnings, and marital status.

Establishment-level data is preferred to firm-level data, since the former allows for observation of flows between establishments within one and the same firm. Nevertheless, we still fail to capture some flows within establishments, since aggregation at establishment level means that some of the turnover of jobs and workers is unaccounted for. Consequently, job gains and losses as well as hirings and separations within the establishment cancel each other out, while only those between establishments are accounted for. Although this is standard occurrence in the literature, it introduces a downward bias in the estimates of flows, as does the fact that we only observe the number of jobs and workers in November each year. The use of annual data means that we will not be able to observe if a job is created and destroyed, or whether a worker is hired and then quits the same establishment between the November of one year to the next. However, the annual flow measure provides a better indicator of permanent job reallocation activity. Moreover, since seasonal employment is low in November,⁵ its selection as the month of observation means that the figures are not much affected by this phenomenon.

The establishment level panel was constructed by linking annual information for observed establishments over time.⁶ By comparing successive years, existing establishments, entries and exits were defined in the following way: if a unit had a new establishment number or if the establishment number was not found during the preceding three years, the establishment was coded as an entry (new establishment); if a previous unit had disappeared and/or did not turn up during the following three years, the unit was considered to be an exit (closure). Units with the same establishment code as in the previous year were coded as existing establishments (survivors). The majority of the establishments, around 87 percent, had

⁴ Distinguishing the births and deaths of establishments from changes in organisational structure, ownership or administrative identifiers may be a problem when CFAR identity number are used, resulting in overestimated job and worker flows, especially due to "false" entries and exits.

⁵ Davis et al. (1996) show that most of the job creation and job destruction captured by quarterly figures reflect establishments-level employment changes that are revised within a year, and Burgess et al. (2000) found that over 20 per cent of employment spells dissolves within a quarter.

⁶Note that we are not following workers over time, the basic observational unit is the establishment.

information for all years they existed. Those establishments where annual information was missing for only one or at most two successive years were treated as continuing. Finally, establishments where annual information were missing for more than 2 successive years (around 2 percent of them) were coded as closures and as new establishments when reappearing.

Some additional restrictions were also applied. Firstly, the analysis considers only establishments that, on average, had at least 5 employees during the establishment's observation period. This restriction might mean that reported job and worker flows will be somewhat biased downwards, since smaller establishments tend to be more volatile with respect to employment. Secondly, employees with several employers have been linked only to their main employer, the employer from whom they have received the highest salary. Thirdly, employees with annual earnings less than one base amount⁷ have been excluded. Due to lack of information in the data we can not distinguish between workers with full-time or part-time jobs. Workers, as such, are given the same weight as long as the earnings restriction is fulfilled.

After data cleaning and exclusions due to restrictions, annual information regarding approximately 110 000 establishments and 3 000 000 employees remained. The data consists of all establishments and their employees in Sweden aged 16–64 years for the period 1986–1989, and 16–65 years for the period 1990–2002. The strength of the data is the combination of detailed employee and establishment information and the very long observation period, covering both a deep economic downturn and a subsequent recovery period. Until now, such detailed data has not been used in analyzing job and worker flows for the Swedish labor market.

4 Job flows and worker flows in Sweden 1986 to 2002

Changes in employment are a result of changes in the supply and demand of labor. Compared to the extensive theoretical and empirical research that has been conducted on labor supply, much less has been related to labor demand. The most

⁷ The amount differs across years. During the period studied, one base amount varied between 23 300 SEK in year 1981 and 37 900 SEK in year 2002. The base amount corresponded in 2002 to slightly less than two monthly average full-time salaries.

important explanation for this is probably the difficulty in obtaining relevant data on establishments and firms. However, since the 1990s, there have been a number of international studies using large linked employer-employee data sets. Studies of several countries have shown that establishments are heterogeneous and that considerable job creation and job destruction flows co-exist at all phases of the business cycle and in all sectors. Some studies have covered both job flows and worker flows. They also find considerable numbers of hirings and separations taking place simultaneously. However, the research in this area is still relatively new.

The study of flows is important in many aspects. Higher rates of job creation and destruction mean larger numbers of workers are compelled to shuffle between jobs and, most likely, there is a greater incidence of unemployment.⁸ For a given net growth rate, higher rates of job creation make it easier for displaced workers and labor market entrants to find employment, and higher rates of job destruction imply less job security for employed persons.

4.1 The concept of job flows

We follow the conventions adopted by Davis & Haltiwanger (1990, 1992) regarding the definitions of job creation and job destruction rates. A job means an employment position occupied by a worker. We use net employment changes as a measure of the business cycle.

Let $E_{(i,t)}$ be employment at establishment *i* in year *t*. Then "job creation" (JC) and "job destruction" (JD) are defined as:

$$JC_{(i,t)} = E_{(i,t)} - E_{(i,t-1)} = \Delta E_{(i,t)} \qquad \text{if } \Delta E_{(i,t)} > 0 \tag{1}$$

$$JD_{(i,t)} = E_{(i,t)} - E_{(i,t-1)} = -\Delta E_{(i,t)} \qquad \text{if } \Delta E_{(i,t)} < 0 \tag{2}$$

The size of the establishment in year t is defined as the average employment of the two years t and t-1. That is

Establishment size =
$$X_{(i,t)} = \frac{1}{2}(E_{(i,t)} + E_{(i,t-1)})$$
 (3)

Dividing JC and JD by the average employment, X, gives us the job creation rate (JCR) and the job destruction rate (JDR).

⁸ Davis et al. (1996), p.11.

JCR is the sum of all jobs created by new establishments (ENTRY) and by expanding establishments (EXP), that is, establishments increasing the number of employees between *t*-1 and *t*, divided by total employment. The JCR_(*t*) is defined as:

$$JCR_{(t)} = \sum_{i \in I_{t}} \left(\frac{\Delta E_{(i,t)}}{X_{(t)}} \right) \text{if } \Delta E_{(i,t)} > 0$$
(4)

$$X_{(t)} = \sum_{i \in I_t} X_{(i,t)}, \ \mathbf{I}_t = \text{all establishments in year } t.$$
(5)

JDR is the sum of all jobs destroyed by closing establishments (EXIT) and by contracting establishments (CONT), that is, establishments reducing their number of employees between *t*-1 and *t*, divided by total employment. The JDR_(*t*) is defined as:

$$JDR_{(t)} = \sum_{i \in I_t} \left(\frac{-\Delta E_{(i,t)}}{X_{(t)}} \right) \text{if } \Delta E_{(i,t)} < 0$$
(6)

The *net employment change* (NET) is the difference between the job creation rate and job destruction rate:

$$NET_{(t)} = JCR_{(t)} - JDR_{(t)}$$
(7)

The *job reallocation rate* (JRR) is the sum of the creation rate and the destruction rate, and is a measure of employment reshuffle across establishments:

$$JRR_{(t)} = JCR_{(t)} + JDR_{(t)}$$
(8)

4.2 The concept of worker flows

The flows of workers are measured as the number of workers moving in and out of establishments, i.e. "hirings" and "separations".⁹ Note that there may be hirings and separations even if the net job change at the establishment is zero.

⁹ We follow the definitions by Burgess et al. (2000).

IFAU - The importance of education for the reallocation of labor

Both individuals employed to replace separations, and those employed in new jobs are defined as hirings. Let $H_{(i,t)}$ denote the number of workers at the establishment in year *t* who did not work there in year *t*-1. The *hiring rate* (HR_(t)) can then be defined as:

$$HR_{(t)} = \sum_{i \in I} \left(\frac{H_{(i,t)}}{X_{(t)}} \right)$$
(9)

Separations may be voluntary (quits) or involuntary (lay-offs). Let $S_{(i,t)}$ denote the number of workers at the establishment in year *t*-1 who do not work there in year *t*. The *separation rate* (SR_(*t*)) can then be defined as:

$$SR_{(t)} = \sum_{i \in I} \left(\frac{S_{(i,t)}}{X_{(t)}} \right)$$
(10)

The difference between the hiring and separation rates is the same as the difference between job creation and job destruction rates, which is the net employment change. That is:

$$JCR_{(t)} - JDR_{(t)} = HR_{(t)} - SR_{(t)} = NET_{(t)}$$
(11)

The worker reallocation rate (WRR_(t)) is defined as:

$$WRR_{(t)} = HR_{(t)} + SR_{(t)}$$
(12)

The relation between worker flows, job flows and changes in employment can be described as follows:

$$WRR_{(t)} \ge JRR_{(t)} \ge NET_{(t)}$$
(13)

The so-called *churning rate* ($ChR_{(r)}$) is the difference between worker flows and job flows. It shows the volume of worker flows in excess of what is needed to meet job flows and can be initiated by either the employer or the employee. The churning rate is defined as:

$$ChR_{(t)} = WRR_{(t)} - JRR_{(t)}$$
(14)

4.3 Flows across groups

So far, jobs and workers have been treated as homogeneous, but job and worker flows are not evenly distributed across groups. The types of workers who lose their jobs at contracting and closing establishments, need not be the same as those who get the new jobs at new and expanding establishments. In order to take this heterogeneity into account, we examine the educational level of workers who get new jobs and of those who lose old ones. We break down the flows further into four groups of educational level: pre-upper secondary, upper secondary, university less than 3 years, and university 3 years or more.

For each group of individuals, the sum of the changes in employment between two consecutive years at an establishment is divided by the total employment for the same group of individuals. The aggregate job creation rate of educational group j is the aggregate increase in jobs for group j for establishments expanding in group j, divided by the number of jobs for group j. The job destruction rate is defined in a similar way by the aggregate reduction of educational group j using the mean of the present and previous size of group j as the denominator.

$$JCR_{(j,t)} = \sum_{i \in I_t} \left(\frac{\Delta E_{(i, j, t)}}{X_{(j, t)}} \right) \text{if } \Delta E_{(i, j, t)} > 0$$
(15)

$$JDR_{(j,t)} = \sum_{i \in I_t} \left(\frac{-\Delta E_{(i, j, t)}}{X_{(j, t)}} \right) if \quad \Delta E_{(i, j, t)} < 0$$
(16)

$$X_{(i,j,t)} = \frac{1}{2} (E_{(i,j,t)} + E_{(i,j,t-I)}), \ X_{(j,t)} = \left(\sum_{i \in I} X_{(i,j,t)}\right)$$
(17)

The hiring rates and separation rates for group j of workers are defined in a similar way:

$$HR_{(j,t)} = \sum_{i \in I} \left(\frac{H_{(i, j, t)}}{X_{(j, t)}} \right)$$
(18)

$$\operatorname{SR}_{(j,t)} = \sum_{i \in I} \left(\frac{S_{(i, j, t)}}{X_{(j, t)}} \right).$$
(19)

IFAU - The importance of education for the reallocation of labor

The rates are presented separately for establishments that enter the market, that increase employment, that reduce employment, that have the same number of people working between two years (stable establishments) and, finally, that exit the market.

5 Estimated flows for the whole population

5.1 Job flows

The net employment change and estimated rates of job creation and job destruction for the period 1986/87 to 2001/02 are shown in *Figure 3*, the rates figures are presented in the appendix, Table A1. Net employment increased by an average of 0.2 percent each year, but with large variations between single years from 2.6 percent 1999/00 to -4.6 percent 1991/92. The net employment change results from large creations and destructions of jobs that occur simultaneously. Each year, on average 10.4 percent of all jobs were created and 10.2 percent were destroyed.

Figure 3. Annual rates of total job destruction and job creation, 1986/87 to 2001/02, percent



Source: The IFAU data base.

Figure 4. Annual rates of job destruction at contracting establishments and exits, and rates of job creation at expanding establishments and entries, 1986/87 to 2001/02, percent



Source: The IFAU data base.

In *Figure 4*, job creation is further divided by new establishments and by expanding establishments, and job destruction is divided by closing establishments and contracting establishments (see Table A1 in the appendix for the figures). Job creation at new establishments (entries), as well as job destruction arising from the closure of establishments (exits) are fairly constant across years. The contribution from continuing establishments to job creation and job destruction is larger than the contribution from new and closing establishments. Three out of four jobs were created or destroyed by continuing establishments, while the remaining 25 percent comes from new and closing establishments.¹⁰

Some results exist for other countries, but a major problem in estimating crosscountry differences is the lack of internationally comparable data. There are, for instance, differences in definitions, sampling intervals and sectors coverage, and we know that job flow rates differ across these factors. Gómez-Salvador et al. (2003) presents job flows of twelve European countries, including Sweden. Since that study includes firms instead of establishments, excludes firms with fewer than 10 employees and only investigates continuing firms, the estimated job flows will

¹⁰ In this study we exclude establishments with less than five employees on average, and we do not distinguish between different types of entries and exits. For a study focusing on entries, see Persson (2004).

be downward biased. Each one of these three factors lowers the job flow rates.¹¹ Nevertheless, Gómez-Salvador et al. presents comparable estimates *across* countries. Of the countries included, Sweden is revealed to have the second highest job reallocation rate.¹²

The correlations between the flows (JCR, JDR, JRR, NET) are presented below in *Table 1*. From the matrix we can see that the correlation between the net employment rate and the job reallocation rate is -0.56, meaning that job reallocation is countercyclical. Job destruction is countercyclical and job creation is procyclical, but the cyclical pattern is more pronounced in the case of job destruction, meaning job destruction is more volatile than job creation. The ratio of the variance in job destruction to the variance in job creation is 3.3.¹³

(JRR,NET):	-0.56				
	(0.024)				
(JCR,NET):	0.69	(JCR _{entry} ,NET):	-0.31	(JCR _{exp.} ,NET):	0.80
	(0.003)		(0.246)		(0.000)
(JDR,NET):	-0.92	(JDR _{exit} ,NET):	-0.29	(JDR _{cont.} ,NET):	-0.94
	(0.000)		(0.267)		(0.000)
Var(JDR)/		Var(JDR _{avit})/		Var(JDR _{cont})/	
	2.2		2.6		2.6
Var(JCR):	3.3	Var(JCR _{entry}):	2.6	Var(JCR _{exp.}):	2.6

Table 1. Pearson correlation coefficients, p-values in parenthesis

These results support Mortensen & Pissarides (1994) model that predicts job reallocation to be countercyclical. According to Garibaldi (1998) the results imply that firing costs (i.e. separation costs) are low in Sweden. Also note that there is no significant cyclical correlation associated with creation due to entries or destruction

¹¹ Gómez-Salvador et al. present an average JCR for Sweden of 8.1 percent and an average JDR of 3.6 percent. Comparable figures for the same time period in our study, excluding entries and exits, are an average JCR of 8.7 percent and an average JDR of 7.0 percent.

¹² The average JRR is 9.8 percent, with the lowest rate of 7.9 percent in Austria and the highest rate of 12.1 percent in Spain. The JRR for Sweden is 11.7 percent.

¹³ Note that an implication of less fluctuation in job creation than in job destruction is that job reallocation is countercyclical. Since JRR is the sum of JC and JD, and NET is the difference between JC and JD, it follows that Cov (JRR,NET) < $0 \rightarrow Var(JD) > Var(JC)$.

due to exits. The cyclical pattern found thereby emerges solely from continuing establishments.

The cyclical behavior of job reallocation, job creation and job destruction has attracted much attention in recent work. While job reallocation in manufacturing has been found to be countercyclical in the U.S. (Davis et al. (1996)), it was found to be non-cyclical in Denmark (Albæk & Sørensen (1998)) and Canada (Baldwin et al. (1998)). Davis & Haltiwanger (1999) present figures for the manufacturing sector in eight countries (U.S., Norway, Canada, U.K., Germany, Netherlands, Denmark and Colombia). Except for in Denmark and Colombia, job destruction was found to be more volatile than job creation, meaning job reallocation was countercyclical.

In this study, we consider how job and worker flows are distributed for the whole Swedish economy. For most countries, information about non-manufacturing industries is limited. One study is that of Foote (1998) who includes alls sectors in Michigan. He finds that job creation fluctuates more than job destruction over time in non-manufacturing sectors. His explanation is that declining sectors have higher variances in job destruction while growing sectors display high variances in job creation. On the other hand, Persson (1999) finds that growing sectors also have higher variances in job destruction, even though the ratio between variance in job destruction to the variance in job creation is smaller in growing sectors. The results found in this study, covering the whole economy and a uniquely long time period, show that job reallocation has been countercyclical despite the fact that net employment has been increasing by an average of 0.2 percent per year. Boeri (1996) presents evidence for 8 countries (U.S., Canada, Denmark, France, Germany, Italy, Norway and Sweden) that covers most or all of the private sector. He finds that apart from in the U.S. manufacturing sector, the job reallocation rate is either acyclical or mildly procyclical.

An explanation for the varying results could be that the time series for most countries covers a relatively short time period and that most studies refer to the manufacturing sector. The results found in this study, covering the whole economy and a long time period, show that the job reallocation has been countercyclical although net employment has been increasing by an average of 0.2 percent per year.

As described later, the results are sensitive to whether workers are treated as homogeneous or heterogeneous. When workers are treated as heterogeneous, different results are received for different groups of workers.

5.2 Worker flows

We have not discussed worker flows so far. The difference between job flows and worker flows consists in the number of workers leaving their jobs and being replaced by other workers. *Figure 5* reveals the figures for worker flows in Sweden during the period 1986/87-2001/02 (Table A2 in the appendix contains the figures for both Figures 5 and 6). Note that the difference between hiring and separation rates for each year in *Figure 5* is the same as the difference between job creation and job destruction rates for each year in *Figure 3*, meaning the net employment change.



Figure 5. Total annual hiring and separation rates, 1986/87 to 2001/02, percent

Source: The IFAU data base.

Average hirings amount to 23.5 percent of total employment each year, and separations amount to 23.3 percent, both somewhat more than twice the job creation and job destruction rates. On average, workers who start and quit in the course of a year (the worker reallocation rate), constitute 46.9 percent of total employment.

The hiring and separation rates at continuing establishments are further divided over expanding, contracting and stable establishments in *Figure 6*. Stable establishments are defined as establishments having the same number of employees in both years. Continuing establishments are responsible for 89 percent of hirings and

separations, while entries and exits are responsible for the remaining 11 percent.¹⁴ The figure reveals that contracting establishments continue to hire workers. Hiring rates in establishments with declining employment are on average 5.6 percent per year. Establishments with expanding employment still lose 6.3 percent of their workers each year.

Figure 6. Annual hiring and separation rates at contracting, expanding and stable establishments, 1986/87 to 2001/02, percent



Source: The IFAU data base.

Studies covering both job and worker flows are scarce. An exception is Albæk & Sørensen (1998) who study worker flows in Danish manufacturing for the period 1980 to 1991. They find a hiring rate of 28.5 percent on average and a separating rate of 28.0 percent. As in Sweden, the worker flows are somewhat more than twice the job flows. The number of hired workers per job created (HR/JCR) and the number of separated workers for each job destroyed (SR/JDR) are both 2.4 in Denmark. In Sweden the numbers are very similar, 2.3 for both measures.

The number of workers who move in and out of establishments (WRR) is much higher than is required to account for the creation or destruction of jobs (JRR) (see *Figure 7*). The churning rate (ChR), defined as the volume of worker flows in excess of what is needed to match job flows, is on average 26 percent. Of the reallocation of workers, only 44 percent is explained by the reallocation of jobs.

¹⁴ Note that hirings and separations at entries and exits are by definition the same as job creation and job destruction at entries and exits.



Figure 7. Annual job reallocation rates (JRR), working reallocation rates (WRR) and churning rates (ChR), 1986/87 to 2001/02, percent

Source: The IFAU data base.

The correlations between the series of worker flows are presented in *Table 2*, where the correlation between worker reallocation and net employment change is positive but insignificant (0.42). However, from *Table 1* we find that job reallocation is countercyclical. While the reallocation of jobs is larger during downturns, this does not hold for reallocation of workers. *Table 2* also reveals that the number of people hired is larger during upturns while the number of people leaving displays no cyclical pattern.

Table 2. Pearson correlation coefficients, p-values in parenthesis

(WRR,NET):	0.42 (0.108)	
(HR,SR):	0.43 (0.093)	
(HR,NET):	0.76 (0.001)	
(SR,NET):	-0.25 (0.345)	
(ChR,NET):	0.61 (0.011)	
Var(SR)/Var(HR):	0.4	

One interpretation is that people are very careful not to leave their jobs during bad times: they only leave when their jobs disappear and they have no choice. Instead, they make the move to quit and find better jobs during upturns. The positive significant correlation between the churning rate and net employment change also supports the interpretation. The share of the reallocation that takes place in excess of what is needed to match job creation and job destruction is larger during upturns than during downturns.

6 Job flows and worker flows for different educational groups

To understand more closely the restructuring process that took place in the economy during the period from 1986/87 to 2001/02 we consider how job and worker flows have been distributed among workers with different educational levels.

In *Figure 8*, net employment changes are shown for the four educational groups: pre-upper secondary, upper secondary, university less than 3 years, and university 3 years or more. The change in employment has not been evenly distributed across educational groups. What is striking is the sharp decline in employment for those with the lowest level of education. While net employment for all workers increased by 0.2 percent each year on average, net employment for those with the least education declined by more than 3 percent each year. For those with a university education, employment increased by more than 2 percent on average, while employment rose by an average of 0.7 percent for workers with upper secondary education. The decline in employment for those with the least education started before the recession, accelerated during the recession and continued to decline during the recovery years.



Figure 8. Net annual employment changes, all establishments, percent

Source: The IFAU data base.

The study by Salvanes & Førre (2003) is the only study we are aware of that evaluates different educational groups in a similar way. They present figures for the Norwegian labor market during the period from 1987 to 1994 for workers with low, medium and high level education in two sectors; manufacturing and finance. Salvanes & Førre find that net employment changes differ over education categories and receive a negative net employment rate of 4 percent for the least educated (in both sectors) and a positive net employment rate for the higher educated of 5 percent in the manufacturing sector and 2 percent in finance. The figures for workers with a medium level of education are in between. Both Norway and Sweden have thus seen a substitution away from a less well-educated to a highly-educated work force.

6.1 Job flows

Figure 9 gives the estimated job creation and job destruction rates for the different educational groups.¹⁵ When comparing job creation and job destruction rates for these four groups it becomes clear that not only were fewer jobs created for those with the least education, but more jobs were also destroyed. During the course of a typical year, 11.4 percent of the jobs were destroyed and 8.0 percent of the jobs were created for those with a pre-upper secondary education. In other words, for

¹⁵ The numbers are presented in Tables A3, A5, A7 and A9 in the appendix.

every one job destroyed for the least educated group, only 0.7 jobs were created (JCR/JDR); for the most highly educated group, 1.3 jobs were created for every one job destroyed. In Norway, the ratios are similar. For the least educated group, 0.7 jobs in manufacturing and 0.8 jobs in the financial sector were created for every one job destroyed. For the most highly educated, the corresponding numbers were found to be 1.4 and 1.1 respectively. The effect of the economic recession for those with a lower level of education is easily seen in *Figure 9*. Job destruction rates increased markedly during the recession years, while job creation rates remained relatively unaffected. This pattern is not found for those with a university educated workers, fewer jobs were destroyed.



Figure 9. Annual job flows for different educational groups, 1986/87 to 2001/02, percent

Source: The IFAU data base.

Even though Sweden and Norway show very similar figures overall, the driving force behind the decreased employment of the least educated workers differs. In Norway job creation rates are higher for highly educated workers, while job destruction rates are more or less the same across education groups. In Sweden, on the other hand, more jobs were destroyed for the less educated than for more highly educated workers.

For the whole labor market, we found job reallocation to be countercyclical. However, when we disaggregate the correlations for different educational groups in *Table 3*, the results vary between groups. Countercyclical behavior is driven only by the lowest educated groups. Job reallocation among those educated to pre-upper secondary level is highly countercyclical, while those educated to upper secondary level also show a countercyclical pattern, although not as strongly. The group having completed less than 3 years of university education has an acyclical pattern, while the group having completed at least 3 years has a procyclical job reallocation pattern.

	Pre-upper	Upper	University	University
	secondary	secondary	< 3 years	\geq 3 years
(JRR,NET):	-0.79 (0.000)	-0.54 (0.031)	-0.17 (0.538)	0.53 (0.034)
(JC,DC):	-0.57 (0.022)	-0.26 (0.337)	0.22 (0.407)	-0.33 (0.208)
(JC,NET):	0.76 (0.001)	0.64 (0.007)	0.54 (0.031)	0.91 (0.000)
(JD,NET):	-0.97 (0.000)	-0.90 (0.000)	-0.70 (0.002)	-0.69 (0.003)
Var(JD)/Var(JC):	6.3	3.2	1.4	0.3
(JC _{entry} , JD _{exit}):	0.21 (0.444)	0.39 (0.131)	0.16 (0.549)	0.42 (0.105)
(JC _{entry} ,NET):	-0.03 (0.924)	-0.20 (0.458)	0.05 (0.849)	0.46 (0.075)
(JD _{exit} ,NET):	-0.65 (0.007)	-0.29 (0.272)	-0.21 (0.434)	0.28 (0.297)
Var(JD _{exit})/				
Var(JC _{entry}):	2.7	1.4	2.5	2.7

Table 3. Pearson correlation coefficients, p-values in parenthesis

This is also seen by the fact that job destruction rates fluctuate much more than job creation rates for those with the least education. The variance of job destruction is more than 6 times that of job creation among those with a pre-upper secondary education. The ratio decreases with education and is less than 1 for those with at least 3 years of university education, meaning that the variance of job creation is higher than that of job destruction. The destruction of jobs held by workers with a pre-upper secondary education fluctuates much more with the business cycle,

revealing a correlation of almost -1 compared to job destruction rates for workers with the highest university education, which reveal a correlation of -0.7.

The differences between groups is down to the fact that the least educated group have had higher job destruction rates during downturns that has not been compensated by higher job creation rates during good times. These results are consistent with Garibaldi (1998) assuming separation costs to be higher for the more highly educated than for the less educated. When this is indeed the case, employers might use recessions for restructuring by separating less well-educated workers and keeping those with higher education.

As previously shown, around 75 percent of the jobs were created/destroyed by existing establishments that expanded/reduced their workforce.¹⁶ The number of jobs that were created/destroyed in new/closed establishments contributes less to the observed net employment changes. This holds for all educational groups and implies that the observed increase in net employment for more highly educated workers has mainly been driven by jobs created and destroyed in continuing firms. (See Tables A3, A5, A7 and A9 in the appendix for figures.)

6.2 Worker flows

Worker flow rates are more than twice as high as job flow rates. In *Figure 10* hiring and separation rates are shown for the four educational groups.¹⁷ Turnover (the sum of HR and SR) has been high for all workers, irrespective of their education. More than 40 percent of all the workers either separated and/or started a new job every year during the observed period. The highest worker flows, 52 percent, are found among those with a university education of less than 3 years. Not surprisingly, the lowest rates, 40 percent, are found among those with the lowest educational level. Corresponding figures for workers with upper secondary education and for workers with a university education of at least 3 years are 49 percent and 47 percent, respectively. Figures for Norway show the same pattern with worker flows of 42 percent for the least educated and over 50 percent for the high-est educated.

Figure 10 clearly shows the different patterns for less and more highly educated workers. While the effect of the economic recession is easily seen for those with pre-upper and upper secondary education, the recession is not directly obvious for those with higher education. The numbers of workers hired per job created were

¹⁶ Comparable figures for Norway are 65-80 percent.

¹⁷ The numbers are presented in Tables A4, A6, A8 and A10 in the appendix.

similar for all educational groups: 2.3 for those educated to pre-upper secondary level, 2.2 for those educated to upper secondary level and 2.4 for those having completed both shorter and longer university education. The numbers of separations per job destroyed are lower for the least educated workers (1.9, 2.2, 2.7 and 2.8 respectively).



Figure 10. Annual worker flows for different educational groups, 1986/87 to 2001/02

Source: The IFAU data base.

Table 4 presents the correlations for different educational groups. According to the table, worker reallocation is found to be acyclical or procyclical for all education groups, even if only the one corresponding to at least 3 years of university education is significant at a level of 5 percent. The correlation is very strong and positive for those with at least 3 years of education, implying that more people are hired and separated during upturns than during downturns. There is a positive correlation between the churning rate and the net employment rate for all groups, meaning that the share of the reallocation that takes place in excess of what

is needed to mach job creation and destruction is larger during upturns than during downturns. The effect is stronger for the most highly educated.

	Pre upper	Upper	University	University
	secondary	secondary	< 3 years	\geq 3 years
(WRR,NET):	0.18	0.43	0.27	0.80
	(0.512)	(0.096)	(0.304)	(0.000)
(ChR,NET):	0.61	0.63	0.39	0.84
	(0.012)	(0.009)	(0.136)	(0.000)
(HR,SR):	0.17	0.40	0.71	0.80
	(0.538)	(0.127)	(0.002)	(0.000)
(HR,NET):	0.72	0.78	0.59	0.91
	(0.002)	(0.000)	(0.017)	(0.000)
(SR,NET):	-0.56	-0.27	-0.15	0.48
	(0.023)	(0.313)	(0.574)	(0.062)
Var(SR)/Var(HR):	0.7	0.4	0.7	0.2

Table 4. Pearson correlation coefficients, p-values in parenthesis

Figure 11. Churning rates for different educational levels



Source: The IFAU data base.

The churning rates (ChR) presented in *Figure 11* show the volume of worker flows in excess of what is needed to match job flows. The rate is given by the difference between the worker reallocation rates (WRR) and the job reallocation

rates (JRR). This rate is high: on average 26 percent for all workers. Much of the mobility that takes place on the labor market can therefore not be explained by the reallocation of jobs. As seen from *Figure 11* (and from Tables A4, A6, A8 and A10 in the appendix), churning rates are higher for those with a higher level of education, on average around 30 per cent for those with a university education. For those with the lowest level of education churning rates are considerably lower, around 20 percent. Thus, mobility that is not motivated by fluctuations in the number of jobs increases with education.

7 Education and the long-run reallocation of labor

During the 1980s and 1990s, the labor market situation for less educated workers deteriorated. At the same time, the supply of highly educated workers increased. The question is whether it is the supply or demand that is the cause behind the change on the labor market.¹⁸ The restructuring during the last two decades could be an effect of a higher demand for the highly educated (highly educated workers have been hired for new jobs that require higher education), or an effect of a greater supply of highly educated individuals (highly educated workers have been hired for the same jobs that were previously filled by less well-educated employees), or a combination of both demand and supply effects.

Previous studies concerned with the restructuring process have found that the demand for highly educated workers has increased on the Swedish labor market. However, they have used methods focusing solely on demand effects and data materials other than linked employer-employee datasets. One explanation for the excess demand is the increased investment in new and more efficient capital equipment, which in turn has increased the relative demand for more highly educated workers through the capital skill complementary mechanism. Another explanation is the increased trade with developing countries.¹⁹

By using employer-employee data for the Norwegian labor market Salvanes & Førre (2003) analyze the employment shift in favor of highly educated workers.

¹⁸ See Salvanes & Førre (2003) for an overview of Norway, which has had a similar development to Sweden.

¹⁹ See Lindquist (2005), Mellander (1999) and Hansson (2000) for studies of skill biased technology change. Hansson also estimates the effect of trade.

They attempt to disentangle the supply and demand effects for different groups by fixing the educational level for old and new cohorts of workers. They then evaluate the pattern of net employment changes for different educational categories within worker cohorts. The authors argue that if higher net employment rates can be found for more highly educated workers from both younger and older cohorts, it will support the explanation that the change in the educational composition has been caused by increased demand for more educated workers. If, on the other hand, only the most highly educated from the youngest cohort obtain the new jobs, this supports the explanation that the change in educational composition has been caused by a greater supply of more highly educated individuals. Salvanes & Førre find that the employment shift in favor of the highly educated in Norway has been caused by both an increased demand and an increased supply of more highly educated workers.

Following Salvanes & Førre we split the data into six two-year cohorts of workers.²⁰ We follow the cohorts from when the employees are 30 years old, assuming that very little education is acquired after the age of 30 (Salvanes & Førre follow the cohorts from the age of 25). We only use data consisting of workers aged 30 or older in 1986, which is the first year in our panel. We then follow the pattern of net employment changes for the four educational groups of different cohorts of workers. Since the two youngest cohorts were not 30 years old in 1986, they are followed from 1989 and 1993 respectively.

The following cohorts were defined: cohort 1, born 1942–1943, aged 43–44 in 1986, cohort 2, born 1946–1947, aged 39–40 in 1986, cohort 3, born 1950–1951, aged 35–36 in 1986, cohort 4, born 1954–1955, aged 31–32 in 1986, cohort 5, born 1958–1959, aged 30–31 in 1989 (followed from 1989), cohort 6, born 1962–1963, aged 30–31 in 1993 (followed from 1993).

For each cohort we assume that the educational level is fixed.²¹ If net employment patterns for different cohorts is similar to the general picture for all cohorts

²⁰ Salvanes & Førre (2003) split their data into four ten-year cohorts of workers. However, in order to control that educational levels are fixed within each cohort, we have chosen narrower intervals.
²¹ The supply of education within the cohorts is fairly stable. For the two oldest cohorts, the proportion

²¹ The supply of education within the cohorts is fairly stable. For the two oldest cohorts, the proportion of those with a university degree is 31 and 33 percent respectively: for the other four cohorts it is about 37 percent. Figure A1 in the appendix shows the educational proportions for each cohort and year.

(with no controls for educational level) it will indicate that the restructuring process that has taken place has been due to an increased demand for highly educated workers. There might well be differences *across* cohorts, for example, that older workers may be more strongly connected to their jobs, but differences in net job creation between educational groups *within* cohorts, means that changes in demand are important.²²

Table 5 presents the average net employment rates for the aggregate that does not control for the supply of education, and for the six cohorts (annual figures are shown in Figure A2 in the appendix).

Table 5. Average net employment rate for cohorts and educational level, $86/87-01/02^{23}$

	All	b. 1942/43	b. 1946/47	b. 1950/51
Pre-upper secondary	-2.90	-1.98	-1.14	-0.79
Upper secondary	0.92	-1.80	-0.84	-0.25
University < 3 years	2.37	-1.22	-0.36	0.21
University ≥ 3 years	2.65	-1.24	-0.52	0.03
	_	b. 1954/55	b. 1958/59	b. 1962/63
		-0.56	-0.44	1.16
		0.04	-0.01	1.24
		0.53	1.14	1.98
		0.69	1.30	2.87

Note: When calculating the average for those born between 1958–1959 and 1962–1963, fewer observations are used, since these cohorts are followed from 1989 and 1993 respectively.

²² In Sweden employment protection is greater for older workers. All workers in regular employment are covered by the Security of Employment Act (often abbreviated as LAS). The seniority rules (last in, first out) stated in LAS, together with more generous calculations of years of service for older workers, gives greater employment protection for older workers. In addition, older workers have a right to longer periods of advance notice than younger workers. Finally, only workers who have reached the age of 40 or over and who have been continuously employed for the last five years are eligible for severance pay. These regulations imply that older workers who quit for a new job will not only be more likely to lose their job if workers are being dismissed (if lay-offs occur) at their new workplace, but they will also lose their right to receive severance pay. From the employer's point of view, higher separation costs are likely to result in a lower demand for older workers, see Calleman (1999) and Jans (2002).

²³ Average net employment rates have also been calculated for shorter time periods, in order to compare the results for different periods of the business cycle. The results from these calculations confirm the overall results that more jobs have been created for highly educated workers (see Table A11 in the appendix for figures of net employment rates for shorter time periods).

The averages for the aggregate show a higher degree of net employment change for the more highly educated than for the less educated workers: a pattern that exists for all cohorts. Consequently, there has been a difference in demand for educational groups that has at least partly been responsible for the difference in net employment change shown earlier in *Figure 8*.

The implication that net employment rates should be higher for highly educated workers within all cohorts can be made operational by comparing Spearman rank correlations. If demand changes are important, rank orderings of net employment rates by level of education in a given cohort should be close to the corresponding rank ordering for the aggregate (all cohorts taken together). The correlation coefficients should be positive and different from zero. As is clear from*Table 6*, the correlations are positive and significant, which imply that the change in the educational composition has at least partly been caused by increased demand for educated workers.

Cohorts	Correlations	No. of obs.		
	(p-values)			
1942–1943	0.50 (0.001)	64		
1946–1947	0.65 (0.001)	64		
1950–1951	0.65 (0.001)	64		
1954–1955	0.52 (0.001)	64		
1958–1959	0.51 (0.001)	52		
1962–1963	0.49 (0.002)	36		

Table 6. Spearman correlation coefficients, p-values in parenthesis

Note: We rank the net employment rates by level of education for each cohort and year. We then test the correlation between the individual cohorts and the aggregate by using Spearman's rank correlation test with the null hypothesis that the correlation is zero.

Although the educational difference in net employment changes holds for each cohort, there is also, as seen from *Table 5*, a difference in the level of net employment changes *across* cohorts. Net employment changes are greater for younger than for older cohorts, and are also greater for the more highly educated than for the less educated. This indicates that the increased supply of highly educated workers might also have been important for the net employment changes on the labor market during the last two decades. However, supply effects are not the only explanation for the observed differences across cohorts. It partly reflects the effect

of older workers leaving the labor market (due to early retirement or retirement), a lower demand for older workers (due to outranged skill and/or higher separation costs), and the effect of entrance into the labor market of younger, more highly educated workers.²⁴

In contrast to Salvanes & Førre, who found a negative net employment change for less educated workers that was of the same order and magnitude across cohorts, we find a larger decrease in net employment for the less educated in older rather than younger cohorts. For the youngest cohort (those born 1962–63) net employment actually increased for less educated workers, by nearly 1.2 percent on average. This might, however, be explained by the establishment investing in new technology, which may expand production in general and increase the demand for all workers, even if demand is relatively higher for more educated workers. Furthermore, the great number of jobs destroyed for less educated workers during the economic downturn in the beginning of the 1990s might have resulted in a reversed effect, that of an increased demand for younger and less educated workers in the years to follow.

All in all, we find support for the idea that the relative employment shift in favor of more highly educated workers has at least been driven by a greater demand for educated workers. The difference in net employment levels across cohorts, i.e. larger net employment changes for highly educated workers in younger rather than in older cohorts, does however suggest that the increased supply of highly educated workers might also have been important for development on the labor market. Nevertheless, the results found for the Swedish labor market give less support for supply effects²⁵ than the results presented by Salvanes & Førre for the Norwegian labor market.

²⁴ For references concerning studies of early retirement among older workers and employers negative attitudes towards recruiting older workers, see National Social Insurance Board (2000).

²⁵ If the increased supply of highly educated workers has been an important cause behind the observed change in the educational composition of the workforce, one might argue that this would be reflected in higher rank orderings for more highly educated workers in younger than in older cohorts. Calculations of rank orderings for highly educated across cohorts are however only found to be higher for younger cohorts than older, in two out of sixteen years studied (1987 and 1990). Rank orderings across cohorts for more highly educated workers are presented in Figure A3 in the appendix.

8 Conclusions

In order to more closely understand the restructuring process that took place in the Swedish economy from 1986 to 2002, we consider how job and worker flows have been distributed both on an aggregate level and across educational levels. We contribute to the previous literature by studying a uniquely long period, a period that covers both downturns and upturns and the whole Swedish economy.

We find that job flows and worker flows are substantial across years, and with huge variations. Further, job reallocation rates are stable both in order and magnitude. Worker flows exhibit a more volatile pattern, but without a trend. We find no support for the idea that job and worker flows have decreased over time, or that flows are lower in Sweden than in other countries, something that has occasionally been stressed in debate and previous literature. Moreover, we confirm previous results that job and worker flows are dominated by flows corresponding to existing establishments, while job creation/hirings in new establishments and job destruction/separations in exiting establishments constitute a minor part.

Analyses of job and worker flows that do not account for the heterogeneity of workers can be very misleading. We contribute to previous studies by decomposing both job and worker flows into educational groups and demonstrate that flows vary by educational level, not only with respect to magnitude and variation but with respect to direction as well.

During the examined period, not only are fewer jobs created for those with the least education, but more jobs are also destroyed. While job destruction rates are found to decrease with educational level, job creation rates follow a more ambiguous pattern. The difference between job and worker flows consists in the number of workers leaving their jobs and being replaced by other others. Worker flows are more than twice as high as job flows for all educational groups and the lowest worker reallocation rate (the sum of hiring and separation rates) is found among the least educated workers. Finally, excess reallocation of workers, i.e. reallocation of workers that is not motivated by fluctuations in the number of jobs, is found to be highest among more highly educated workers.

Some previous studies have found job reallocation to be countercyclical, suggesting that downturns are periods of restructuring the establishment. Our results confirm those previous studies on an aggregate level and support the model by Mortensen & Pissarides (1994) predicting job reallocation to be countercyclical. According to Garibaldi (1998) this countercyclical pattern implies that firing costs (i.e. separation costs) are low. However, when examining the correlations between

different educational groups, the countercyclical behavior was only found among those with low educational level. For workers with the highest education, job reallocation rates were even found to be procyclical. Employers seem to use recessions for restructuring by separating less educated workers and by keeping more highly educated workers. The results are consistent with Garibaldi (1998) if assuming separation costs to be greater for the more highly educated, than for the less educated workers.

While the reallocation of jobs is found to be larger during downturns on the aggregate level, this does not hold for reallocation of workers. Instead worker reallocation exhibits an acyclical pattern. The number of people hired is larger during upturns while the number of people leaving displays no cyclical pattern. One interpretation is that people are very careful not to leave their jobs during bad times and only leave when they have to, preferring to quit during upturns to find better jobs. There are some differences between educational groups. Worker reallocation for the most highly educated shows a strong procyclical pattern due to both more hirings and separations during up-turns. The least educated workers, on the other hand, have significantly more separations during downturns.

Following Salvanes & Førre (2003), we also examine the long-run reallocation of labor for different educational groups. To decide whether the reallocation of workers has been driven by an increased demand for more highly educated workers, an increased supply, or a combination of both, net employment rates for different cohorts were analyzed more closely. Our results indicate that the relative employment shift in favor of more highly educated workers has been driven by a higher demand for educated workers, but also that the increased supply of more highly educated workers might have been important.

References

Abowd J M, P Corbel & F Kramarz (1999) "The entry and exit of workers and the growth of employment: An analysis of French establishments", *Review of Economics and Statistics*, Vol. 81, No. 2, pp. 170-87.

Abowd J M & F Kramarz (1999) "The analysis of labor markets using matched employer-employee data", in: Ashenfelter O & D Card (eds.), *The Handbook of Labour Economics*, Vol. 3B, Chapter 40, Amsterdam, North-Holland.

Albæk K & B E Sørensen (1998) "Worker flows and job flows in Danish manufacturing, 1980-91", *The Economic Journal*, Vol. 108, November, pp. 1750-1771.

Andersson F (2003) *Causes and labour market consequences of producer heterogeneity*. PhD. Thesis, Dissertation series 2003:1, Institute for Labour Market Policy Evaluation (IFAU), Uppsala University, Uppsala.

Andersson L, O Gustafsson & L Lundberg (1998) "Structural change, competition and job turnover in the Swedish manufacturing industry 1964-96", Working paper No. 148, Swedish Trade Union Institute for Economic Research (FIEF).

Arai M & F Heyman (2000) "Permanent and temporary labour: job and worker flows in Sweden, 1989-1998." Working paper No. 71, Konjunkturinstitutet.

Baldwin J, T Dunne & J Haltiwanger (1998) "A comparison of job creation and job destruction in Canada and the United States", *The Review of Economics and Statistics*, Vol. 80, No. 3, pp. 347-356.

Belzil C (1997) "Job creation and destruction. Worker reallocation and wages", Working paper 97-04, Centre for Labour Market and Social Research, University of Aarhus and the Aarhus School of Business.

Boeri T (1996) "Is job turnover counter cyclical?", *Journal of Labor Economics*, Vol. 14, No. 4, pp. 603-625.

Burgess S, J Lane & D Stevens (2000) "Job flows, worker flows, and churning", *Journal of Labor Economics*, Vol. 18, No. 3, pp. 473-502.

Calleman K (1999) *Turordning vid uppsägning*. [Seniority rules for lay-offs] PhD. Thesis, Department of Law, Umeå University, Umeå.

Davis S J & J Haltiwanger (1990) "Gross job creation and destruction: microeconomics evidence and macroeconomics implications", in: Blanchard O & S Fisher (eds.), *NBER Macroeconomics Annual*. The MIT Press, Cambridge, MA.

Davis S J & J Haltiwanger (1992) "Gross job creation, gross job destruction and employment reallocation", *The Quarterly Journal of Economics*, Vol. 107, No 3, pp. 819-864.

Davis S & J Haltiwanger (1999) "Gross job flows", in: Ashenfelter O & D Card (eds.), *The Handbook of Labour Economics*, Vol. 3B, Chapter 41, Amsterdam, North-Holland.

Davis S J, J Haltiwanger & S Schuh (1996), *Job Creation and Destruction*. The MIT press, Cambridge, Massachusetts & London, England.

Davis S J, R J Farberman & J Haltiwanger (2005) "The flow approach to labor markets: New data sources, micro-macro links and the recent downturn" Discussion Paper No. 1639, The Institute for the Study of Labor (IZA).

Foote C L (1998) "Trend employment growth and the bunching of job creation and destruction", *Quarterly Journal of Economics*, Vol. 113, No. 3, pp. 809-934.

Garibaldi P (1998) "Job flow dynamics and firing restrictions", *European Economic Review*, Vol. 42, No. 2, pp. 245-275.

Gómez-Salvador R, J Messina & G Vallanti (2003) "Gross job flows and institutions in European countries", mimeo.

Hamermesh D S, W H J Hassink & J C Van Ours (1994) "Job turnover and labor turnover: A taxonomy of employment dynamics", Research Memorandum 1994-50, Vrije Universiteit, Amsterdam. Hansson P (2000) "Relative demand for skills in Swedish manufacturing: Technology or trade?", *Review of International Economics*, Vol. 8, No. 3, pp. 533-555.

Jans A-C (2002) *Notifications and Job Losses on the Swedish Labour Market*. PhD. Thesis, Dissertation Series, No. 54, Swedish Institute for Social Research (SOFI), Stockholm University, Stockholm.

Lane J, A Isaac & D Stevens (1996) "Firm heterogeneity and worker turnover", *Review of Industrial Organization*, Vol. 11, No. 3, pp. 275-291.

Lindquist M (2005) "Capital-skill complementarity and inequality in Sweden", Working paper 2/2005, Swedish Institute for Social Research (SOFI), Stockholm University, Stockholm.

Mellander E (1999) "The multi-dimensional nature of labor demand and skillbiased technical change", Working Paper 1999:9, Institute for Labour Market Policy Evaluation (IFAU), Uppsala.

Mortensen D T & C A Pissarides (1994) "Job creation and destruction in the theory of unemployment", *Review of Economic Studies*, Vol. 61, No. 3, pp. 397-415.

National Social Insurance Board (2000) "Social insurance in Sweden 2000. After 55 – Welfare, work and leisure", Riksförsäkringsverket (RFV), Stockholm.

Nordström Skans O, P-A Edin & B Holmlund (2006) "Wage dispersion between and within plants: Sweden 1985-2000", Working Paper 2006:9, Institute for Labour Market Policy Evaluation (IFAU), Uppsala.

Persson H (1999) "Job flows and worker flows in Sweden 1986-1995", in *Essays* on Labour Demand and Career Mobility. PhD. Thesis, Dissertation Series, No. 40, Swedish Institute for Social Research (SOFI), Stockholm University, Stockholm.

Persson H (2004) "The survival and growth of new establishments in Sweden, 1987-1995", *Small Business Economics*, Vol. 23, No. 5, pp. 423-440.

Salvanes K & S E Førre (2003) "Effects on Employment of Trade and Technical Change: Evidence from Norway", *Economica*, Vol. 70, No. 2, pp. 293-329.

APPENDIX

		JCR				JDR		
	Net	Entry	Exp.	Total	Exit	Cont.	Total	JRR
1986/87	2.07	2.97	8.49	11.46	3.34	6.05	9.39	20.85
1987/88	1.40	2.73	7.62	10.35	2.33	6.61	8.94	19.29
1988/89	1.19	2.93	7.03	9.96	2.05	6.71	8.77	18.73
1989/90	-0.53	2.73	6.43	9.15	1.79	7.89	9.68	18.83
1990/91	-1.28	2.64	6.48	9.12	2.24	8.16	10.40	19.52
1991/92	-4.64	3.57	6.62	10.19	2.93	11.90	14.83	25.02
1992/93	-4.53	2.58	6.25	8.83	2.88	10.48	13.36	22.19
1993/94	-0.16	3.00	8.02	11.02	2.69	8.50	11.18	22.20
1994/95	1.77	2.51	8.45	10.96	2.25	6.94	9.19	20.15
1995/96	0.71	2.74	7.27	10.01	1.87	7.44	9.30	19.31
1996/97	-0.50	2.37	7.26	9.63	2.14	7.99	10.13	19.76
1997/98	2.46	2.27	8.63	10.90	1.90	6.54	8.44	19.34
1998/99	2.30	2.79	8.85	11.64	2.44	6.89	9.34	20.98
1999/00	2.63	2.91	9.15	12.06	2.55	6.88	9.43	21.49
2000/01	0.90	2.63	8.36	10.99	2.35	7.74	10.09	21.08
2001/02	-0.66	2.52	7.89	10.41	3.37	7.70	11.07	21.47
Average	0.20	2.74	7.67	10.42	2.44	7.78	10.22	20.64

Table A1. Job reallocation and its components, 1986/87 to 2001/02, percent

		Hiring rates					Separation rates						
	Net	Entry	Exp.	Cont.	Stable	Total	Exit	Exp.	Cont.	Stable	Total	WRR	ChR
1986/87	2.07	2.97	15.71	5.46	1.24	25.39	3.34	7.22	11.51	1.24	23.31	48.70	27.84
1987/88	1.40	2.73	15.02	5.97	1.29	25.01	2.33	7.40	12.58	1.29	23.60	48.61	29.32
1988/89	1.19	2.93	14.50	6.70	1.36	25.50	2.05	7.48	13.42	1.36	24.30	49.80	31.07
1989/90	-0.53	2.73	13.01	7.47	1.38	24.59	1.79	6.59	15.36	1.38	25.12	49.71	30.88
1990/91	-1.28	2.64	11.56	6.00	1.17	21.36	2.24	5.08	14.15	1.17	22.64	44.00	24.49
1991/92	-4.64	3.57	10.67	5.40	0.95	20.58	2.93	4.06	17.29	0.95	25.22	45.81	20.79
1992/93	-4.53	2.58	10.07	5.34	0.94	18.93	2.88	3.82	15.83	0.94	23.46	42.39	20.20
1993/94	-0.16	3.00	13.16	4.93	1.05	22.15	2.69	5.14	13.43	1.05	22.31	44.45	22.25
1994/95	1.77	2.51	14.32	5.00	1.20	23.02	2.25	5.87	11.94	1.20	21.25	44.28	24.12
1995/96	0.71	2.74	12.69	4.90	1.17	21.50	1.87	5.41	12.34	1.17	20.79	42.29	22.97
1996/97	-0.50	2.37	12.42	5.14	1.13	21.05	2.14	5.16	13.13	1.13	21.55	42.60	22.84
1997/98	2.46	2.27	15.46	4.79	1.22	23.73	1.90	6.83	11.33	1.22	21.28	45.01	25.67
1998/99	2.30	2.79	16.07	5.41	1.28	25.55	2.44	7.23	12.30	1.28	23.25	48.81	27.84
1999/00	2.63	2.91	16.92	5.88	1.36	27.07	2.55	7.77	12.76	1.36	24.44	51.51	30.02
2000/01	0.90	2.63	16.53	6.29	1.37	26.83	2.35	8.17	14.03	1.37	25.93	52.75	31.67
2001/02	-0.66	2.52	15.25	5.34	1.27	24.38	3.37	7.36	13.03	1.27	25.04	49.42	27.95
Average	0.20	2.74	13.96	5.63	1.21	23.54	2.44	6.29	13.40	1.21	23.34	46.88	26.24

 Table A2.
 Worker reallocation and its components, 1986/87 to 2001/02, percent

		JCR JDR							
	Net	Entry	Exp.	Total	Exit	Cont.	Total	Stable	JRR
1986/87	-1.37	2.86	6.08	8.94	3.29	6.84	10.31	-0.18	19.26
1987/88	-1.49	2.61	5.31	7.93	2.25	6.99	9.42	-0.18	17.35
1988/89	-1.73	2.98	5.33	8.31	2.27	7.62	10.04	-0.15	18.35
1989/90	-1.80	2.76	5.36	8.12	2.04	7.82	9.92	-0.06	18.04
1990/91	-5.18	2.51	4.70	7.21	2.62	9.55	12.39	-0.22	19.60
1991/92	-11.54	2.85	3.85	6.70	3.34	14.49	18.24	-0.41	24.94
1992/93	-8.20	2.27	4.73	7.00	3.49	11.51	15.20	-0.20	22.20
1993/94	-4.90	2.49	5.31	7.80	3.10	9.33	12.70	-0.27	20.50
1994/95	-2.20	2.11	5.70	7.82	2.48	7.32	10.02	-0.22	17.84
1995/96	-3.30	2.00	4.88	6.89	1.96	7.96	10.19	-0.26	17.07
1996/97	-3.38	2.07	5.20	7.28	2.23	8.22	10.66	-0.21	17.93
1997/98	-0.11	2.11	6.72	8.84	2.11	6.69	8.95	-0.15	17.79
1998/99	0.43	2.75	7.64	10.38	2.45	7.44	9.96	-0.07	20.34
1999/00	-1.62	2.43	6.66	9.09	2.66	7.83	10.71	-0.21	19.79
2000/01	-2.74	2.22	6.22	8.43	2.46	8.53	11.17	-0.19	19.61
2001/02	-5.05	2.28	5.43	7.71	3.42	9.09	12.75	-0.24	20.46
Average	-3.39	2.46	5.57	8.03	2.64	8.58	11.41	-0.20	19.44

Table A3. Job reallocation and its components, 1986 to 2002; workers educated to pre-upper secondary level, percent

			Hiring rates					Sep	aration	ates			
	Net	Entry	Exp.	Cont.	Stable	Total	Exit	Exp.	Cont.	Stable	Total	WRR	ChR
1986/87	-1.37	2.86	12.29	3.79	1.00	19.94	3.29	6.21	10.63	1.18	21.31	41.25	21.99
1987/88	-1.49	2.61	11.63	4.41	1.07	19.72	2.25	6.31	11.40	1.25	21.21	40.93	23.58
1988/89	-1.73	2.98	11.71	4.95	1.13	20.77	2.27	6.38	12.57	1.28	22.49	43.26	22.58
1989/90	-1.80	2.76	10.55	5.78	1.17	20.25	2.04	5.18	13.60	1.23	22.05	42.31	22.88
1990/91	-5.18	2.51	9.07	4.20	0.92	16.69	2.62	4.37	13.75	1.14	21.88	38.57	18.97
1991/92	-11.54	2.85	8.12	3.37	0.69	15.02	3.34	4.26	17.86	1.09	26.56	41.58	16.63
1992/93	-8.21	2.27	7.96	3.34	0.67	14.24	3.49	3.24	14.85	0.87	22.44	36.68	14.48
1993/94	-4.90	2.49	10.05	3.11	0.76	16.41	3.10	4.74	12.44	1.03	21.31	37.72	17.22
1994/95	-2.20	2.11	10.84	3.10	0.90	16.96	2.48	5.14	10.42	1.12	19.16	36.12	18.28
1995/96	-3.30	2.00	9.32	3.12	0.82	15.27	1.96	4.44	11.09	1.08	18.57	33.84	16.77
1996/97	-3.38	2.07	9.50	3.53	0.87	15.98	2.23	4.30	11.75	1.07	19.36	35.34	17.41
1997/98	-0.11	2.11	12.35	3.81	1.04	19.31	2.11	5.62	10.50	1.19	19.42	38.74	20.95
1998/99	0.43	2.75	13.69	4.42	1.17	22.03	2.45	6.06	11.86	1.24	21.60	43.62	23.28
1999/00	-1.62	2.43	13.53	4.18	1.12	21.26	2.66	6.88	12.01	1.33	22.88	44.13	24.34
2000/01	-2.74	2.22	12.97	4.85	1.21	21.25	2.46	6.75	13.38	1.40	23.99	45.24	25.63
2001/02	-5.05	2.28	12.35	4.21	1.11	19.95	3.42	6.92	13.31	1.35	24.99	44.94	24.48
Aver.	-3.39	2.46	11.00	4.01	0.98	18.44	2.64	5.42	12.59	1.18	21.83	40.27	20.59

Table A4. Worker reallocation and its components, 1986 to 2002; workers educated to pre-upper secondary level, percent

		JCR JDR							
	Net	Entry	Exp.	Total	Exit	Cont.	Total	Stable	JRR
1986/87	3.86	3.32	9.96	13.36	3.59	5.91	9.50	0.08	22.86
1987/88	2.83	2.87	8.90	11.85	2.51	6.52	9.03	0.08	20.88
1988/89	1.83	3.17	7.58	10.79	2.09	6.87	8.96	0.04	19.75
1989/90	1.72	2.98	7.70	10.84	1.86	7.27	9.13	0.16	19.97
1990/91	-1.58	2.73	6.85	9.60	2.32	8.86	11.18	0.03	20.79
1991/92	-3.91	4.22	7.58	11.92	3.13	12.70	15.83	0.11	27.74
1992/93	-4.86	2.80	6.63	9.46	3.02	11.30	14.32	0.02	23.78
1993/94	0.17	3.38	8.81	12.24	2.83	9.24	12.08	0.06	24.32
1994/95	2.55	2.92	9.23	12.19	2.38	7.26	9.64	0.04	21.83
1995/96	0.61	2.90	7.69	10.61	1.99	8.01	10.00	0.02	20.61
1996/97	-0.17	2.57	7.70	10.29	2.19	8.27	10.46	0.02	20.75
1997/98	2.39	2.47	8.95	11.44	2.04	7.01	9.05	0.02	20.49
1998/99	1.98	2.82	8.79	11.61	2.39	7.23	9.63	-0.01	21.24
1999/00	3.39	3.06	9.67	12.78	2.56	6.83	9.39	0.05	22.18
2000/01	1.44	2.77	8.71	11.54	2.35	7.74	10.10	0.06	21.64
2001/02	-1.26	2.53	7.67	10.21	3.31	8.16	11.47	0.01	21.68
Average	0.69	2.97	8.28	11.30	2.54	8.07	10.61	0.05	21.91

 Table A5. Job reallocation and its components, 1986 to 2002; workers educated to upper secondary level, percent

	_	Hiring rates					Separation rates						
	Net	Entry	Exp.	Cont.	Stable	Total	Exit	Exp.	Cont.	Stable	Total	WRR	ChR
1986/87	3.86	3.32	17.74	6.25	1.42	28.72	3.59	7.77	12.16	1.34	24.86	53.59	30.73
1987/88	2.83	2.87	16.95	6.82	1.48	28.11	2.51	8.04	13.34	1.39	25.28	53.39	32.51
1988/89	1.83	3.17	15.80	7.48	1.53	27.97	2.09	8.22	14.35	1.48	26.14	54.12	34.36
1989/90	1.72	2.98	14.62	8.36	1.60	27.56	1.86	6.92	15.63	1.43	25.85	53.41	33.44
1990/91	-1.58	2.73	12.26	6.63	1.34	22.96	2.32	5.41	15.49	1.31	24.53	47.49	26.70
1991/92	-3.91	4.22	11.85	6.26	1.16	23.49	3.13	4.26	18.96	1.05	27.40	50.89	23.15
1992/93	-4.86	2.80	10.47	5.47	1.05	19.80	3.02	3.84	16.77	1.03	24.66	44.46	20.68
1993/94	0.17	3.38	14.15	5.13	1.20	23.86	2.83	5.34	14.37	1.15	23.70	47.56	23.24
1994/95	2.55	2.92	15.25	5.00	1.36	24.54	2.38	6.02	12.26	1.32	21.99	46.52	24.69
1995/96	0.61	2.90	13.16	4.94	1.32	22.32	1.99	5.47	12.95	1.30	21.71	44.03	23.42
1996/97	-0.17	2.57	12.88	5.13	1.29	21.87	2.19	5.18	13.41	1.27	22.04	43.91	23.15
1997/98	2.39	2.47	15.83	4.84	1.39	24.52	2.04	6.87	11.85	1.37	22.14	46.66	26.17
1998/99	1.98	2.82	16.08	5.37	1.40	25.66	2.39	7.28	12.59	1.41	23.68	49.33	28.09
1999/00	3.39	3.06	17.31	5.78	1.50	27.65	2.56	7.64	12.61	1.45	24.26	51.91	29.73
2000/01	1.44	2.77	16.60	6.45	1.57	27.39	2.35	7.89	14.20	1.51	25.95	53.34	31.71
2001/02	-1.26	2.53	15.14	5.56	1.42	24.65	3.31	7.46	13.72	1.41	25.91	50.56	28.87
Average	0.69	2.97	14.75	5.97	1.38	25.07	2.54	6.48	14.04	1.33	24.38	49.45	27.54

Table A6. Worker reallocation and its components, 1986 to 2002; workers educated to upper secondary level, percent

	_		JCR			JDR			
	Net	Entry	Exp.	Total	Exit	Cont.	Total	Stable	JRR
1986/87	3.42	2.73	9.17	11.95	3.39	5.14	8.53	0.05	20.48
1987/88	2.71	2.82	8.41	11.23	2.27	6.23	8.51	-0.01	19.74
1988/89	2.04	2.51	6.98	9.50	1.74	5.72	7.47	0.01	16.97
1989/90	3.45	2.55	7.74	10.41	1.30	5.67	6.96	0.12	17.38
1990/91	1.00	2.43	6.84	9.33	1.73	6.60	8.33	0.06	17.66
1991/92	1.16	3.67	7.99	11.93	2.24	8.53	10.77	0.26	22.70
1992/93	-0.90	2.73	7.31	10.22	2.30	8.82	11.12	0.18	21.34
1993/94	4.05	3.11	9.77	13.05	2.32	6.68	9.00	0.17	22.05
1994/95	3.36	2.29	10.03	12.41	1.96	7.08	9.04	0.08	21.45
1995/96	2.94	3.15	8.12	11.38	1.60	6.84	8.44	0.12	19.83
1996/97	0.14	2.45	8.11	10.63	2.06	8.43	10.49	0.07	21.12
1997/98	3.33	2.14	9.37	11.54	1.68	6.52	8.21	0.02	19.75
1998/99	3.47	3.17	9.56	12.74	2.44	6.83	9.27	0.01	22.01
1999/00	1.45	2.98	8.85	11.82	2.58	7.71	10.37	-0.08	22.20
2000/01	1.94	2.73	8.88	11.69	2.30	7.44	9.75	0.08	21.43
2001/02	0.32	2.52	8.69	11.26	3.64	7.30	10.94	0.05	22.20
Average	2.12	2.75	8.49	11.32	2.22	6.97	9.20	0.08	20.52

Table A7. Job reallocation and its components, 1986 to 2002; workers having received less than 3 years of university education, percent

			н	liring rat	es		Separation rates						
	Net	Entry	Exp.	Cont.	Stable	Total	Exit	Exp.	Cont.	Stable	Total	WRR	ChR
1986/87	3.42	2.73	17.44	6.52	1.28	27.97	3.39	8.28	11.66	1.22	24.55	52.51	32.03
1987/88	2.71	2.82	16.68	6.26	1.28	27.03	2.27	8.27	12.49	1.28	24.32	51.36	31.61
1988/89	2.04	2.51	15.02	7.15	1.34	26.02	1.74	8.03	12.87	1.34	23.99	50.01	33.04
1989/90	3.45	2.55	14.67	8.76	1.44	27.42	1.30	6.93	14.42	1.32	23.97	51.39	34.01
1990/91	1.00	2.43	12.91	7.52	1.29	24.16	1.73	6.08	14.12	1.23	23.16	47.32	29.66
1991/92	1.16	3.67	12.49	8.10	1.25	25.52	2.24	4.49	16.64	0.99	24.36	49.87	27.18
1992/93	-0.9	2.73	11.30	7.01	1.12	22.17	2.30	3.99	15.83	0.94	23.06	45.23	23.89
1993/94	4.05	3.11	15.49	7.21	1.30	27.12	2.32	5.72	13.88	1.14	23.06	50.18	28.13
1994/95	3.36	2.29	17.25	6.89	1.40	27.85	1.96	7.22	13.98	1.32	24.48	52.33	30.88
1995/96	2.94	3.15	15.00	6.47	1.43	26.05	1.60	6.88	13.31	1.32	23.11	49.16	29.33
1996/97	0.14	2.45	14.62	6.77	1.32	25.17	2.06	6.52	15.19	1.25	25.02	50.19	29.07
1997/98	3.33	2.14	18.03	6.05	1.34	27.56	1.68	8.66	12.57	1.32	24.23	51.80	32.05
1998/99	3.47	3.17	18.71	6.84	1.41	30.13	2.44	9.15	13.67	1.39	26.65	56.78	34.77
1999/00	1.45	2.98	17.65	6.71	1.33	28.66	2.58	8.81	14.41	1.41	27.22	55.88	33.68
2000/01	1.94	2.73	18.55	7.14	1.46	29.88	2.30	9.67	14.58	1.38	27.94	57.82	36.38
2001/02	0.32	2.52	17.70	6.51	1.45	28.18	3.64	9.00	13.81	1.41	27.86	56.05	33.85
Aver.	2.12	2.75	15.85	6.99	1.34	26.93	2.22	7.36	13.97	1.27	24.81	51.74	31.22

Table A8. Worker reallocation and its components, 1986 to 2002; workers having received less than 3 years of university education, percent

			JCR			JDR			
	Net	Entry	Exp.	Total	Exit	Cont.	Total	Stable	JRR
1986/87	1.41	2.10	7.22	9.32	2.70	5.19	7.92	-0.02	17.24
1987/88	-0.18	2.27	6.12	8.39	1.89	6.66	8.58	-0.03	16.97
1988/89	1.10	2.19	6.45	8.64	1.68	5.82	7.54	-0.05	16.18
1989/90	0.96	2.01	6.40	8.48	1.22	6.31	7.53	0.07	16.01
1990/91	1.93	2.62	6.68	9.33	1.69	5.71	7.40	0.03	16.73
1991/92	-1.27	2.33	6.06	8.39	2.02	7.62	9.63	-0.03	18.03
1992/93	-2.06	2.05	5.89	7.94	1.88	8.12	10.00	0.00	17.93
1993/94	2.90	2.35	8.04	10.50	1.84	5.75	7.59	0.10	18.09
1994/95	2.68	1.88	7.60	9.52	1.78	5.06	6.84	0.04	16.35
1995/96	3.37	2.72	7.54	10.38	1.59	5.43	7.01	0.11	17.39
1996/97	1.64	2.00	7.36	9.48	1.85	6.00	7.85	0.12	17.33
1997/98	4.35	1.91	8.62	10.59	1.37	4.88	6.25	0.06	16.84
1998/99	4.52	2.36	9.51	11.98	2.59	4.87	7.46	0.11	19.43
1999/00	5.62	2.88	10.11	13.11	2.34	5.15	7.48	0.11	20.59
2000/01	5.11	2.56	10.06	12.78	2.21	5.46	7.67	0.16	20.45
2001/02	3.89	2.68	9.81	12.62	3.21	5.53	8.74	0.14	21.36
Average	2.25	2.31	7.72	10.09	1.99	5.85	7.84	0.06	17.93

Table A9. Job reallocation and its components, 1986 to 2002; workers having received at least 3 years of university education, percent

		Hiring rates					Separation rates						
	Net	Entry	Exp.	Cont.	Stable	Total	Exit	Exp.	Cont.	Stable	Total	WRR	ChR
1986/87	1.41	2.10	14.13	5.28	1.02	22.53	2.70	6.91	10.48	1.04	21.13	43.66	26.43
1987/88	-0.18	2.27	12.96	5.46	1.03	21.73	1.89	6.84	12.12	1.06	21.91	43.64	26.67
1988/89	1.10	2.19	13.27	5.81	1.10	22.37	1.68	6.82	11.62	1.15	21.27	43.64	27.46
1989/90	0.96	2.01	12.30	7.25	1.13	22.70	1.22	5.90	13.56	1.06	21.74	44.45	28.44
1990/91	1.93	2.62	12.73	6.61	1.01	22.98	1.69	6.05	12.32	0.98	21.05	44.02	27.29
1991/92	-1.27	2.33	11.15	6.77	0.81	21.06	2.02	5.09	14.39	0.84	22.33	43.40	25.34
1992/93	-2.06	2.05	10.38	6.27	0.75	19.45	1.88	4.49	14.39	0.76	21.51	40.97	23.03
1993/94	2.90	2.35	13.77	6.37	0.97	23.47	1.84	5.73	12.12	0.87	20.57	44.04	25.95
1994/95	2.68	1.88	13.95	6.66	1.01	23.50	1.78	6.35	11.73	0.97	20.82	44.32	27.97
1995/96	3.37	2.72	13.97	6.36	1.04	24.09	1.59	6.42	11.79	0.92	20.72	44.82	27.43
1996/97	1.64	2.00	13.74	6.82	0.99	23.55	1.85	6.38	12.82	0.87	21.91	45.46	28.13
1997/98	4.35	1.91	16.95	5.85	1.01	25.72	1.37	8.33	10.73	0.95	21.38	47.10	30.26
1998/99	4.52	2.36	17.80	6.57	1.14	27.87	2.59	8.29	11.44	1.03	23.35	51.22	31.78
1999/00	5.62	2.88	18.43	7.13	1.22	29.66	2.34	8.32	12.27	1.10	24.03	53.69	33.10
2000/01	5.11	2.56	19.64	7.76	1.15	31.11	2.21	9.58	13.22	0.99	26.00	57.11	36.66
2001/02	3.89	2.68	18.36	6.36	1.18	28.58	3.21	8.55	11.89	1.05	24.70	53.28	31.92
Average	2.25	2.31	14.60	6.46	1.04	24.40	1.99	6.88	12.31	0.98	22.15	46.55	28.62

Table A10. Worker reallocation and its components, 1986 to 2002; workers having received at least 3 years of university education, percent

Period 86/87–88/89	All	Born	Born	Born	Born	Born	Born
		1942/43	1946/47	1950/51	1954/55	1958/59	1962/63
Pre-upper secondary	0,34	0,69	0,78	0,90	0,35	-	-
Upper secondary	3,80	0,52	0,91	1,28	0,95	-	-
University < 3 years	3,62	0,48	1,00	1,32	0,87	-	-
University \geq 3 years	2,33	0,09	0,40	0,86	1,26	-	-
Period 89/90–92/93	All	Born	Born	Born	Born	Born	Born
		1942/43	1946/47	1950/51	1954/55	1958/59	1962/63
Pre-upper secondary	-6,37	-2,25	-1,94	-2,22	-2,39	-2,63	-
Upper secondary	-2,01	-1,76	-1,42	-1,16	-1,21	-1,83	-
University < 3 years	1,43	-0,98	-0,14	0,30	0,74	1,01	-
University \geq 3 years	0,19	-0,94	-0,59	-0,23	-0,01	0,26	-
Period 93/94–01/02	All	Born	Born	Born	Born	Born	Born
		1942/43	1946/47	1950/51	1954/55	1958/59	1962/63
Pre-upper secondary	-2,44	-2,75	-1,43	-0,72	-0,05	0,53	1,16
Upper secondary	1,26	-2,59	-1,16	-0,36	0,30	0,80	1,24
University < 3 years	2,37	-1,89	-0,91	-0,19	0,32	1,19	1,98
University \geq 3 years	3,85	-1,82	-0,80	-0,12	0,82	1,76	2,87

 Table A11. Average net employment rate for cohorts and educational levels, different periods



Figure A1. The supply of education within cohorts

Figure A2. Annual net employment rates for all cohorts, and for six cohorts with respect to education, 1986/87–2001/02



IFAU - The importance of education for the reallocation of labor



Figure A3. Rankorder correlations for different cohorts having received less than 3 years and 3 years or more of university education for the period 1986/87–2001/02

Publication series published by the Institute for Labour Market Policy Evaluation (IFAU) – latest issues

Rapporter/Reports

- **2007:1** Lundin Daniela "Subventionerade anställningar för unga en uppföljning av allmänt anställningsstöd för 20–24-åringar"
- 2007:2 Lundin Daniela, Eva Mörk & Björn Öckert "Maxtaxan inom barnomsorgen påverkar den hur mycket föräldrar arbetar?"
- **2007:3** Bergemann Annette & Gerard van den Berg "Effekterna av aktiv arbetsmarknadspolitik för kvinnor i Europa – en översikt"
- **2007:4** Junestav Malin "Socialförsäkringssystemet och arbetsmarknaden politiska idéer, sociala normer och institutionell förändring en historik"
- 2007:5 Andersson Christian "Lärartäthet, lärarkvalitet och arbetsmarknaden för lärare"
- 2007:6 Larsson Laura & Caroline Runeson "Effekten av sänkt sjukpenning för arbetslösa"
- **2007:7** Stenberg Anders "Hur påverkar gymnasialt komvux löneinkomster och vidare studier?
- **2007:8** "Forslund Anders & Kerstin Johansson "Lediga jobb, arbetssökande och anställningar den svenska matchningsfunktionen"
- **2007:9** Kennerberg Louise "Hur förändras kvinnors och mäns arbetssituation när de får barn?"
- 2007:10 Nordin Martin "Invandrares avkastning på utbildning i Sverige"
- **2007:11** Johansson Mats & Katarina Katz "Underutnyttjad utbildning och lönegapet mellan kvinnor och män"
- **2007:12** Gartell Marie, Ann-Christin Jans & Helena Persson "Utbildningens betydelse för flöden på arbetsmarknaden"

Working Papers

- **2007:1** de Luna Xavier & Per Johansson "Matching estimators for the effect of a treatment on survival times"
- **2007:2** Lundin Daniela, Eva Mörk & Björn Öckert "Do reduced child care prices make parents work more?"
- **2007:3** Bergemann Annette & Gerard van den Berg "Active labor market policy effects for women in Europe a survey"

- **2007:4** Andersson Christian "Teacher density and student achievement in Swedish compulsory schools"
- **2007:5** Andersson Christian & Nina Waldenström "Teacher supply and the market for teachers"
- **2007:6** Andersson Christian & Nina Waldenström "Teacher certification and student achievement in Swedish compulsory schools"
- **2007:7** van den Berg Gerard, Maarten Lindeboom & Marta López "Inequality in individual mortality and economic conditions earlier in life"
- **2007:8** Larsson Laura & Caroline Runeson "Moral hazard among the sick and unemployed: evidence from a Swedish social insurance reform"
- **2007:9** Stenberg Anders "Does adult education at upper secondary level influence annual wage earnings?"
- **2007:10** van den Berg Gerard "An economic analysis of exclusion restrictions for instrumental variable estimation"
- **2007:11** Forslund Anders & Kerstin Johansson "Random and stock-flow models of labour market matching Swedish evidence"
- 2007:12 Nordin Martin "Immigrants' return to schooling in Sweden"
- **2007:13** Johansson Mats & Katarina Katz "Wage differences between women and men in Sweden the impact of skill mismatch"
- **2007:14** Gartell Marie, Ann-Christin Jans & Helena Persson "The importance of education for the reallocation of labor: evidence from Swedish linked employer-employee data 1986–2002"

Dissertation Series

- 2006:1 Hägglund Pathric "Natural and classical experiments in Swedish labour market policy"
- 2006:2 Savvidou Eleni "Technology, human capital and labor demand"
- **2006:3** Söderström Martin "Evaluating institutional changes in education and wage policy"
- 2006:4 Lagerström Jonas "Discrimination, sickness absence, and labor market policy"
- **2006:5** Johansson Kerstin "Empirical essays on labor-force participation, matching, and trade"